Author’s response to reviews

Title: Consortium on Vulnerability to Externalising Disorders and Addictions (cVEDA): A developmental cohort study protocol

Authors:

Eesha Sharma (eesha.250@gmail.com)
Nilakshi Vaidya (nilakshivaidya@ymail.com)
Udita Iyengar (udita.iyengar@kcl.ac.uk)
Yuning Zhang (yuning.zhang@kcl.ac.uk)
Bharath Holla (hollabharath@gmail.com)
Meera Purushottam (meera.purushottam@gmail.com)
Amit Chakrabarti (amit.chakrabarti@gov.in)
Gwen Fernandes (gwen.fernandes@bristol.ac.uk)
Jon Heron (jon.Heron@bristol.ac.uk)
Matthew Hickman (matthew.Hickman@bristol.ac.uk)
Sylvane Desrivieres (sylvane.desrivieres@kcl.ac.uk)
Kamakshi Kartik (kamakshi.kartik@gmail.com)
Preeti Jacob (preetijacob@gmail.com)
Madhavi Rangaswamy (madwee@gmail.com)
Rose Bharath (drrosedawnbharath@gmail.com)
Gareth Barker (gareth.barker@kcl.ac.uk)
Dimitri Orfanos (dimitri.papadopoulos@cea.fr)
Chirag Ahuja (chiragkahuja@gmail.com)
Pratima Murthy (pratimamurthy@gmail.com)
Sanjeev Jain (docsanjeev.jain@gmail.com)
It is unclear whether the map used in Figure 4 is copyrighted and whether permission is required for its use. We would recommend that a map is used which can be confirmed is not under copyright or permission has been obtained for its use.

RESPONSE: I have changed the image to one that is free to use. Have clarified this in the Figure legend as well.